

ŗ

Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL085202

Key Points:

- Proxies for hydroclimate and catchment ecology show insolation-driven trends, with higher δD values during the LGM similar to outputs from climate models
- Reduced summer evaporation during cold episodes increased water availability
- Increased summer moisture caused higher lake levels at 31–30, 28–26, and 17–14 kyr BP coinciding with northern hemispheric rapid climate events

Supporting Information:

Supporting Information S1

Correspondence to: B. Aichner.

bernhard.aichner@gmx.de

Citation:

Aichner, B., Makhmudov, Z., Rajabov, I., Zhang, Q., Pausata, F. S. R., Werner, M., et al. (2019). Hydroclimate in the Pamirs was driven by changes in precipitation-evaporation seasonality since the last glacial period. *Geophysical Research Letters*, *46*, 13,972–13,983. https://doi.org/10.1029/2019GL085202

Received 29 AUG 2019 Accepted 11 NOV 2019 Accepted article online 15 NOV 2019 Published online 3 DEC 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Hydroclimate in the Pamirs Was Driven by Changes in Precipitation-Evaporation Seasonality Since the Last Glacial Period

Bernhard Aichner^{1,2}, Zafar Makhmudov³, Iljomjon Rajabov³, Qiong Zhang⁴, Francesco S. R. Pausata⁵, Martin Werner⁶, Liv Heinecke¹, Marie L. Kuessner⁷, Sarah J. Feakins⁸, Dirk Sachse⁹, and Steffen Mischke¹⁰

¹Institute of Earth and Environmental Science, University of Potsdam, Potsdam-Golm, Germany, ²Department of Ecosystem Research, Leibniz-Institute of Freshwater Ecology and Inland Fisheries, Berlin, Germany, ³Climate Change and Ozone Center, State Administration for Hydrometeorology of the Republic of Tajikistan, Dushanbe, Tajikistan, ⁴Department of Physical Geography and Bolin Centre for Climate Research, Stockholm University, Stockholm, Sweden, ⁵ESCER Centre, Department of Earth and Atmospheric Sciences, University of Quebec in Montreal, Montreal, Quebec, Canada, ⁶Department of Climate Sciences, Alfred Wegener Institute, Bremerhaven, Germany, ⁷Équipe de Géochimie des Enveloppes Externes, Institut de Physique du Globe de Paris, Paris, France, ⁸Department of Earth Sciences, University of Southern California, Los Angeles, CA, USA, ⁹Helmholtz Centre Potsdam, German Research Centre for Geosciences GFZ, Section 4.6 Geomorphology, Potsdam, Germany, ¹⁰Faculty of Earth Sciences, University of Iceland, Reykjavík, Iceland

Abstract The Central Asian Pamir Mountains (Pamirs) are a high-altitude region sensitive to climatic change, with only few paleoclimatic records available. To examine the glacial-interglacial hydrological changes in the region, we analyzed the geochemical parameters of a 31-kyr record from Lake Karakul and performed a set of experiments with climate models to interpret the results. δD values of terrestrial biomarkers showed insolation-driven trends reflecting major shifts of water vapor sources. For aquatic biomarkers, positive δD shifts driven by changes in precipitation seasonality were observed at ca. 31–30, 28–26, and 17–14 kyr BP. Multiproxy paleoecological data and modelling results suggest that increased water availability, induced by decreased summer evaporation, triggered higher lake levels during those episodes, possibly synchronous to northern hemispheric rapid climate events. We conclude that seasonal changes in precipitation-evaporation balance significantly influenced the hydrological state of a large waterbody such as Lake Karakul, while annual precipitation amount and inflows remained fairly constant.

Plain Language Summary Lakes in arid Central Asia are particularly susceptible to the rise and fall of lake levels as a consequence of climatic changes. To evaluate drivers behind this phenomenon, we developed a record of humidity and lake levels throughout the last 31,000 years from a high-altitude lake in the Pamir Mountains. Herefore, we combined hydrological and ecological reconstructions with climate model experiments. Results show that neither the enhanced inflow by melting glaciers nor the significantly increased precipitation amount was responsible for higher lake levels during the studied interval. Instead, reduced summer evaporation during cold episodes was the major trigger for lake transgressions. These fluctuations were driven by changes in radiative forcing (i.e., insolation and hence temperature change) as a consequence of changes in the Earth's orbit around the Sun. As such, our results suggest that a significant impact on lake levels in arid regions is also to be expected by the current anthropogenically driven global warming.

1. Introduction

The Pamirs are located at the intersection of atmospheric influences from the Asian monsoon system, the midlatitude westerly winds (westerlies), and the wintertime Siberian anticyclone (Aizen et al., 2001). They are characterized by a pronounced gradient of decreasing winter and spring precipitation from west to east (Miehe et al., 2001; Williams & Konovalov, 2008). Past changes in hydroclimate in this region were likely to experience shifts in the extent of those large-scale atmospheric circulation systems (Caves Rugenstein & Chamberlain, 2018; Zhang et al., 2017). There is a consensus that the Asian monsoon is driven by insolation on a precessional scale, with increasing intensity since the late glacial and a maximum in the early to mid-Holocene (Herzschuh, 2006; Wang et al., 2017). However, mechanisms and the local and regional expression

of short-term rapid climate events, as well as the coupling of temperature and hydrological changes, are still under debate (Peterse et al., 2011; Sarkar et al., 2015). Principally, the north-south movement of the jetstream over Central Asia is directly linked to the monsoonal front in Southeast Asia (Schiemann et al., 2009) and a strong Asian summer monsoon is potentially coupled to an enhanced winter monsoon in Central Asia (Li et al., 2013). However, the local to regional responses in this vast area are not fully understood (Chen et al., 2010), and it is unclear if and how far the Indian branch of the monsoons, after a weak phase during H1 (Deplazes et al., 2014; Tierney et al., 2016), had extended into Central Asia during the early Holocene.

While westerlies dominated the local hydroclimate during the Holocene based on proxy evidence (Aichner et al., 2015; Yan et al., 2019), studies on glacial to interglacial climate in the region are mainly restricted to climate model simulations (Li et al., 2013; Zhang et al., 2018). Pre-Holocene climatic records from Central Asia are limited to speleothems from Ton Cave (Uzbekistan) and Kesang Cave (Tien Shan, China) and loess sections from western Tajikistan, all located several hundreds of kilometers to the Northwest and East of the Central Pamirs, respectively (Cai et al., 2017; Cheng et al., 2016; Haggi et al., 2019; Figure 1).

Lake Karakul is a closed-basin lake (388 km²) in the eastern Pamirs, Tajikistan (approx. 39°N and 73°25′E) where it fills a graben structure (Strecker et al., 1995). The average mean annual precipitation (MAP) within the shielded Karakul Basin is <100 mm, while the MAP at the higher altitudes is often >1,000 mm, mainly due to enhanced winter precipitation. Mean January, July, and annual temperatures are -18.1, 8.5, and -4.0° C, respectively (Williams & Konovalov, 2008). The lake catchment extends over 4,464 km² and reaches up to peaks at 6,780 m a.s.l. from the lake altitude of 3,915 m a.s.l. (Figure 1). However, paleoshorelines as high as 4,120 m a.s.l. indicate that the lake responded sensitively to hydrological changes, with the last high-stand, 35 m above the present lake level, occurring at 15 kyr BP, synchronous with glacial advances in the catchment (Komatsu & Tsukamoto, 2015).

Recently, a lacustrine archive with a robust chronology spanning a glacial-interglacial cycle was obtained from Lake Karakul (Mischke et al., 2017). Inorganic geochemical data, sedimentological, and paleoecological proxies from this record showed large-scale shifts, possibly driven by insolation, over the past 28 kyr (Heinecke, Epp, et al., 2017; Heinecke, Mischke, et al., 2017; Heinecke et al., 2018). Some of those proxies were interpreted with respect to past lake level changes, but hydrological inferences were tentative.

To decipher past hydrological conditions in Lake Karakul and its catchment in more detail, we analyzed geochemical proxies including the hydrogen isotopic signatures of both terrestrial and aquatic biomarkers whose values are directly linked to the hydrological cycle. Results were placed in context to a set of experiments using two atmospheric models with embedded isotope modules CAM3-iso (Noone & Sturm, 2010) and ECHAM5-wiso (Werner et al., 2011), as well as the climate model EC-Earth (Hazeleger et al., 2010; Hazeleger et al., 2012). With this coupled proxy-model approach and with consideration of prior ecological and geochemical evidence, we aim to (I) evaluate regional hydrological responses to large-scale climatic shifts over a glacial-interglacial cycle; and (II) decipher potential drivers behind inferred lake-level changes.

2. Materials and Methods

A detailed description of methodology is given in supporting information Text S1. Briefly, a composite agemodel was built for two parallel sediment cores (KK12-1 and -2) taken in 12-m water depth (39.01760°N and 73.53275°E; Mischke et al., 2017). For organic geochemical analysis, samples from core KK12-2 were extracted with an organic solvent and split into fractions by using silica-gel columns. Concentrations, carbon, and hydrogen isotopic values of compounds in the aliphatic and ketone fraction were obtained by gas chromatography and isotope ratio mass spectrometry under application of standard measures for quality assurance. The aquatic to terrestrial ratio (P_{aq}) was calculated according to Ficken et al. (2000): $P_{aq} = (nC_{23} + nC_{25})/(nC_{23} + nC_{25} + nC_{29} + nC_{31})$. $\Delta_{terr-aq}$ was calculated as $1,000 \times [(\delta D-terr + 1,000)/(\delta D-aq + 1,000) -$ 1] using the arithmetic mean between $\delta D-C_{23}$ and $\delta D-C_{25}$ as δD -aq and between $\delta D-C_{29}$ and $\delta D-C_{31}$ as δD -terr. Elemental data were derived from XRF-scanning and inorganic isotopic values obtained by methods described in Heinecke, Epp, et al. (2017), Heinecke, Mischke, et al.(2017) and—if measured at core KK12-1 —were transferred to the new composite age-model.





Figure 1. Lake Karakul and its catchment with coring site (modified after Taft et al., 2014). Seasonal climatic data (blue bar, mean monthly precipitation; red curve, mean monthly temperature) from Karakul station (yellow dot; 1934–1990) in comparison to Sarytash (ca. 60 km north; 1934–2000) and Fedchenko glacier (ca. 65 km west; 1936–1994; Williams & Konovalov, 2008). Insert: A = Kesang Cave. B = Ton Cave. C = Darai Kalon loess section.

To simulate climate parameters for the Last Glacial Maximum (LGM; 21 kyr BP), Heinrich 1 (H1; 16 kyr BP), mid-Holocene (MH; 6 kyr BP), and Present (PD; 1970–2000), the isotope-enabled global climate models CAM3-iso and ECHAM5-wiso, and the ocean-atmosphere coupled model EC-Earth were used (Collins et al., 2006; Hazeleger et al., 2012; Roeckner et al., 2003; Roeckner et al., 2006; Sturm et al., 2010; Werner et al., 2011). They are characterized by $2.8 \times 2.8^{\circ}$ (CAM3-iso) and $1.125 \times 1.125^{\circ}$ (ECHAM5-wiso, EC-Earth) grid size. For interpretation of windspeed, the output derived from CAM3-iso was used, because with largest grid size, it integrates the widest regional signal of all models. In contrast, ECHAM5-wiso and EC-Earth were considered to reflect the wider lake catchment and preferred for subsequent interpretations of local hydrological and climatic parameters.

3. Results and Discussion

3.1. Biomarker Abundance and Composition-Assessing Organic Matter Sources

The organic matter in most sections was dominated by aquatic sources (P_{aq} value ~0.8; Ficken et al., 2000; Figure 2i), evidenced by visible weakly decomposed macrophyte remains and high concentrations of nC_{23} - and nC_{25} -alkanes (approx. 1.5–12 µg/g d.w.) in the sediment extract, especially after ca. 7 kyr BP (compiled stack-plot of all measured parameters in the supporting information Figure S1). The relatively high δ^{13} C signatures of those compounds (approx. -15%) provide further support for a submerged aquatic macrophyte and macroalgae origin (Aichner et al., 2010) with Potamogetonaceae and Characeae as the predominant families (Heinecke, Epp, et al., 2017). Less abundant nC_{29} - and nC_{31} -alkanes (0–2.5 µg/g d.w.) with δ^{13} C values of approximately -30% are considered to reflect the sparse alpine C₃ vegetation in the catchment today, possibly with minor contribution from biomarkers derived from species thriving in small patches of wetlands around the lake. Terrestrial compounds were present in lower concentrations (<0.5 µg/g) during most parts of the glacial as expected during glacial expansion and cooling, and a more persisting ice-cover on the lake. Three episodes revealed increased terrestrial plant wax concentrations during the late Marine Isotope Stage (MIS) 3 and early MIS 2 at ca. 31–30 and 28–26 kyr BP, and during the late glacial at 17–14 kyr BP (Figure S1).



Figure 2. Top: (a) Winter insolation (Berger & Loutre, 1991). (b, c) Karakul (KK) long-chain leaf-wax δ D values in comparison to Ti areas from XRF scanning and (f, g) δ^{18} O data from Ton and Kesang Cave speleothems (Cheng et al., 2016) and (h) δ D values from Darai Kalon loess section (Haggi et al., 2019). (e) $\Delta_{terr-aq}$: Offset between δ D values of aquatic and terrestrial *n*-alkanes in comparison to (d) winter westerlies index (WWI; Li et al., 2013). Bottom: Proxies indicative for lake level and vegetation changes. (i) P_{aq} : Proxy for aquatic macrophytes (Ficken et al., 2000). (j) Summer insolation. (k) Reconstruction of 15 kyr BP lake level highstand (Komatsu & Tsukamoto, 2015). (l) PCA axis 2, indicating lake level (Heinecke, Mischke, et al., 2017) based on inorganic proxies measured at core KK12-1, adjusted to the new composite age-model for both parallel cores KK12-1 and -2, as published in Mischke et al. (2017). (m, n) Inferred salinity changes based on Sr/Ca ratios and alkenone abundances. (o) δ^{18} O of aragonite (yellow curve; Heinecke, Epp, et al., 2017; Heinecke, Mischke, et al., 2017). High-resolution glacial δ^{18} O values of aragonite (red curve; this study). (r) Dominant aquatic species at the coring location. Shaded intervals indicate periods of increasing summer humidity and higher lake level.

Those intervals also have a low aquatic to terrestrial ratio (P_{aq} value ~0.3, Figure 2i) and were further characterized by the occurrence of unsaturated compounds (nC_{23} , nC_{25} , and nC_{27} alkenes; Figure 2q). While the exact origin of those molecules remains unclear, a source from phytoplankton such as eustigmatophytes and chlorophytes and possibly from insects is likely (Aichner et al., 2012; van Bree et al., 2014; Zhang et al., 2015). Occurrence of *n*-alkenes correlate with increased counts of Chironomid remains at 28–26 and 15–14 kyr BP (Heinecke et al., 2018). Conversely, a much broader range of compounds than observed in the Karakul sediments would be expected with insects as a significant precursor of those compounds (Chikaraishi et al., 2012; van Bree et al., 2014). Long-chain C_{37} alkenones derived from phytoplankton (de Leeuw et al., 1979) were abundant between ca. 17–5 kyr BP (10–200 ng/g d.w; supporting information Figure S1). For the overall trend of organic matter input, bromine content derived from XRF scanning is used as an estimate (Figure S1), because many organic molecules contain carbon bound bromine, which is released after decomposition (Mayer et al., 1981).

3.2. Biomarker Stable Isotopes as Recorders of Past Local Hydrology and Moisture Sources

Inorganic and organic geochemical proxies showed distinct glacial-interglacial trends over the past 31 kyrs (Figures 2 and S1). Since nC_{29} and nC_{31} alkanes were mainly derived from the sparse alpine meadows around the lake, their δD values reflect a precipitation signal recorded during the vegetation period (i.e., summer, with a possible contribution of a winter signal by early spring snow-melt) within the lower altitude catchment (Figure 2c). In contrast, we interpret aquatic nC_{23} and nC_{25} alkane δD values to record a lake water isotopic signal that integrates over the whole year, including meltwater derived from higher altitudes (Figure 2p). Because of the interpretations for δD -aq (winter + summer signal) and δD -terr (summer signal), we infer the offset (Δ) between them as a proxy for winter precipitation and thus being driven by the location and intensity of the westerly flow during winter (Figure 2e).

The higher δD -terr values during the glacial and lower values during the Holocene are in agreement with speleothem $\delta^{18}O$ data from Ton Cave and Kesang Cave 8 (Cai et al., 2017; Cheng et al., 2016; Figure 1) and leaf wax δD values from a loess section in western Tajikistan (Haggi et al., 2019), highlighting the transregional relevance of this signal (Figures 2f, 2g, and 2h). Seasonal isotopic variability of precipitation in inner Asia is driven by temperature (Araguas-Araguas et al., 1998; Yao et al., 2013). However, the higher δD -terr values during the glacial are unlikely to reflect a higher temperature. Higher titanium counts during the glacial (Figure 2b) could indicate drier conditions; however, the large isotopic amplitude within the δD -terr data (approx. 60%) cannot be solely explained by further aridification of a lower lake basin already characterized by present day MAP of <100 mm. Hence, we suggest that the shift is driven by a large-scale change of water vapor sources between glacial and Holocene summers, while higher titanium abundance in this case indicates intensified winds rather than further aridification during colder conditions.

In agreement with this interpretation of titanium counts, the CAM3-iso model simulated higher winter windspeeds for LGM and H1 compared to the Holocene (Figures 3 and 4a). This is related to an LGM eastward shift of North Atlantic storm tracks that resulted from atmospheric responses to the expansion of ice-sheet and sea-ice (Kageyama, D'Andrea, et al., 1999; Kageyama, Valdes, et al., 1999; Laine et al., 2009). Wind fields further illustrate the dominance of the southern branch of westerlies during winter and spring and resulting southwesterly wind in the Pamirs over the studied time intervals (Figure S2a). In contrast, in summers the region is more dominated by the northern branch with northern winds (in agreement with computed back trajectories by Yan et al., 2019), which is an additional explanation for the lower precipitation compared to winter and spring.

CAM3-iso also showed that most of the summer moisture in the study area is derived from continental recycling (supporting information Figure S3). Thus, the relevant driver behind isotopic variability during summers is the balance between (more enriched) local and continental moisture sources and those which have been transported farther (more depleted). The shift toward more depleted δD -precip values at the glacial-interglacial boundary could therefore be interpreted as a strengthening of either the monsoon (increased transport from Arabian Sea) or of the westerlies (increased vapor transport from North Atlantic; Figure S3).

Adding to this complexity is the observation that monsoonal strength and westerly air-masses are closely linked to each other. An intensified summer monsoon in East Asia is associated to a jet-stream located over





Figure 3. Comparison of outputs of CAM3-iso, ECHAM5-wiso, and EC-Earth models. Average values in Table S1. Temperature and precipitation data in comparison to present day instrumental data (colored lines; Williams & Konovalov, 2008). δD data in comparison to data from the Online Isotope in Precipitation Calculator (OIPC; red line; Bowen & Revenaugh, 2003). RH = relative humidity; PME = precipitation minus evaporation.





Figure 4. Anomaly plots LGM versus MH for summer (JJA; left) and winter (DJF; right) for (a) windspeed (850-hPa sigma level, derived from CAM3-iso), (b) precipitation, and (c) precipitation minus evaporation (PME) both derived from EC-Earth. Study area indicated by arrow.

the Pamirs (Schiemann et al., 2009), resulting in stronger summer westerlies. Based on a 150-kyr long simulation, Li et al. (2013) suggested that phases of increased northern hemispheric insolation and more intense Asian monsoon are potentially also accompanied by an increase of winter precipitation in Central Asia, depending on latitude. They proposed that low winter insolation and resulting high latitude cooling decrease the northern hemispheric temperature gradient, (low winter westerlies index WWI; Rossby, 1939), which in turn weakens the northern westerly streams but enforces the southern branch. It is difficult to assess the consequences of those findings with regard to precipitation amount changes in our study area, as it is located at the proposed latitudinal boundary of 40°N. However, our suggested indicator for westerlies intensity (Δ_{terr} - $_{aq}$) is in phase with winter insolation (Berger & Loutre, 1991) and the WWI (Li et al., 2013), supporting this interpretation (Figures 2a, 2d, and 2e). Weak winter westerlies between ca. 17–7 kyr BP in our study area correspond to findings from the Tien Shan, where similar conditions have been reconstructed for the late glacial and early Holocene, based on grain size analysis of a loess section (Jia et al., 2018).

A late glacial and early Holocene intensification of the Asian monsoons is well established (Hou et al., 2017; Rao et al., 2016; Wang et al., 2005), but it is unclear if the Pamirs were significantly influenced by moisture derived from the Indian summer monsoon, even though this is suggested by the most negative δD-terr values in our Karakul record between ca. 12–7 kyr BP. The co-occurrence of strengthening of monsoons and westerlies in Central Asia makes it difficult to determine the precise source of moisture for those time intervals. The modelling data suggest that precipitation over the region is controlled by the westerlies all around the year and for the studied time periods (Figures S2a and S2b). On the contrary, based on the model outputs, it also becomes clear that the western Pamirs are affected in a significantly stronger way by the incoming summer north-westerlies than our study area in the eastern Pamirs (Figure S2a), resulting in the observed E-W gradient of spring precipitation (Figure S2b) and of summer PME (precipitation-evaporation balance; Figure S2c). This illustrates that the Karakul region is situated in a transitional area in terms of precipitation amount, water balance, and wind directions for all simulated time intervals.

We conclude that the changes of atmospheric circulation patterns during the late glacial to Holocene transition were characterized by a decreased winter westerly flow and a shift from local to distant vapor sources during summers. The stronger summer monsoon over southeast Asia and India during that time significantly influenced the regional hydroclimate in the Pamirs—either directly by the monsoon itself or more likely by influencing the pathways and intensity of the westerlies.

3.3. Lake Level Highstands During Cold Episodes Caused by Wetter Summers

Several independent proxies provide evidence that Lake Karakul reacted sensitively to changes of local hydrological and climatic conditions. Based on OSL dating of paleoshorelines, Komatsu and Tsukamoto (2015) identified several lake transgressions, the last starting before 19 kyr BP and forming a shoreline 35 m higher than present around 15 kyr BP (Figure 2k). In addition, Heinecke, Mischke, et al. (2017) interpreted higher values on the second axis of a principal component analysis of geochemical parameters, described by high TIC content and Fe/Mn ratios, to reflect glacial and late glacial lake-level highstands (Figure 2l). In our sedimentary record, we further interpret an increasing Sr/Ca ratio after 14 kyr BP as indicator for increasing salinity in a shrinking lake (Dissard et al., 2010; Yan & Wünnemann, 2014; Figure 2m). The interval 14–10 kyr BP was also characterized by decreasing proportional abundance of tetra-unsaturated alkenones $C_{37:4}$ (Figure 2n), which was found to be related to higher salinities based on studies of surface sediment samples from lakes on the Tibetan Plateau (Liu et al., 2008; Liu et al., 2011) even though other factors such as temperature, nutrient availability, and changing producers might complicate this relationship.

Taken together, all these proxies suggest rising lake level in the late glacial (peaking around 15 kyr BP) and episodically also in the last glacial period (between ca. 31–30 and 28–26 kyr BP). These phases are accompanied by ecological changes within the lake (shift in dominant submerged plants at coring location; occurrence of biomarkers such as alkenones and C_{23} , C_{25} , and C_{27} *n*-alkenes, giving evidence for a change in algal communities) as well as around the lake (change from desert to steppe vegetation; Heinecke et al., 2018). Unsaturated midchain compounds were discovered in many Tibetan Plateau lakes. Their precise source is still unknown, but analysis of surface sediments of Lake Donggi Cona revealed that they mainly occur in greater depths (Aichner et al., 2012). At present, *Chara* species occur rather in deeper parts of Lake Karakul, while Potamogetonaceae thrives down to a maximum depth of approximately 14–15 m (Heinecke, Epp, et al., 2017; Heinecke, Mischke, et al., 2017). The massive occurrence of weakly decomposed Potamogetonaceae remains after ca. 7 kyr BP, and synchronous increase of nC_{23} and nC_{25} concentrations in sediments were therefore most likely induced by the combined effects of a falling lake level, change in lake water chemistry (disappearance of alkenones), and resulting changes in light penetration to the ground, leading to pronounced propagation of submerged macrophytes at the coring position (12-m water depth at present).

We interpret the enhanced terrestrial contribution (lower P_{aq}) at ca. 31–30, 28–26, and 17–14 kyr BP as a key to understand the occurrences of lake-level highstands during those intervals. Since vegetation in hyperarid

environments reacts sensitively to slight changes in humidity, we suggest that enhanced summer humidity during those intervals caused the higher influx of terrestrial *n*-alkanes, which occurred synchronous to higher pollen counts and a change from alpine desert to steppe vegetation (Heinecke et al., 2018). While enhanced discharge from glacial meltwater is less likely (glaciers in the Pamirs were still either at a maximum or even expanding in the late MIS 3/early MIS 2 and in the late glacial; Dortch et al., 2013; Komatsu & Tsukamoto, 2015), an explanation of the higher lake levels are shifts in the hydrological balance toward higher precipitation and/or lower lake-water evaporation.

3.4. Changing Seasonality of the PME as Major Driver of Hydrological Changes

The simulated differences in mean annual δD -precip between the LGM and PD were approximately +11‰, according to the outputs of CAM3-iso and ECHAM5-wiso models (Table S1). This is in range of the observed offset in δD -aq (representing annual δD ; Figure 2p) in the Karakul record. The +60‰ shift in δD -aq between 17 and 14 kyr BP, however, is not reflected in CAM3-iso model simulations for Heinrich event 1 (H1). Similar shifts around 30, 24, and 16 kyr BP were observed in speleothem $\delta^{18}O$ records from East Asia and were hypothesized to be related to events H3 to H1 (Wang et al., 2001). Through a modeling study, Pausata et al. (2011) showed that the weakening of the Indian summer monsoon during those events leads to higher δD -precip over India and transport of anomalously D-enriched vapor to East Asia. This explanation refers to a source isotopic signature of summer precipitation and cannot be directly transferred to our study area, neither mechanistically nor with respect to the Karakul isotopic record, which lacks such a pattern in δD -terr. We hypothesize that episodes of increased summer humidity and higher lake levels at Lake Karakul were indeed a local response to northern hemispheric rapid climate events, but with more regional causes for positive excursions in aquatic isotopes.

A change of nC_{23} sources (Potamogetonaceae to Characeae) could explain at least part of the large isotopic change, although limited data and contradictory results of previous studies make a full evaluation of the potential effect difficult (Liu, Yang, Cao, Leng, & Liu, 2018; Liu, Yang, Cao, & Liu, 2018). A synchronous shift toward higher values was also observed in δ^{18} O of aragonite crystals (δ^{18} O-ara; Heinecke, Epp, et al., 2017; Heinecke, Mischke, et al., 2017; Figure 2o). Between ca. 14–12 kyr BP, these two proxies for isotopic composition of lake water drift apart, visible in a sharp drop of δ D-aq in contrast to only slightly decreasing δ^{18} O-ara. A shrinking lake level and increase of salinity could be a reason for this pattern, as salinity is implicated to influence isotopic fractionation of aquatic macrophyte biomarkers (Aichner et al., 2017). However, this effect is not yet fully understood, and it needs to be noted that δ D-aq follows the major drop of δ D-terr; hence this most likely reflects the source signal of precipitation.

We propose that changes in seasonality of precipitation provide the most likely mechanism for positive shifts in δ D-aq even though it cannot be excluded that changes in aquatic macrophyte sources and salinity contributed to the signal. Isotope-enabled models and data from the Online Isotope Precipitation Calculator (Bowen & Revenaugh, 2003) reveal a strong seasonal amplitude of δ D values, with summer precipitation up to 100% more positive than winter precipitation (Figure 3). Hence, increased summer precipitation and/or lower influx from winter precipitation best explain the +60% shifts in δ D-aq around 31–30, 28–26, and 17–14 kyr BP (Figure 2p).

The models give further support that regional hydrology at Lake Karakul was mainly driven by changes in seasonality of effective humidity. They all indicate slightly wetter summers (increased precipitation and relative humidity) and more arid and windier winters during LGM and H1 compared to the Holocene (Figures 3 and 4b), while the annual precipitation amounts remained relatively constant (Table S1). The EC-Earth model suggests that strongly decreased evaporation during cold episodes is a relevant influencing factor behind the increased summer humidity as simulated by the models (Figure 3; Table S1). Proxy evidence points toward late glacial lake transgression starting during cold episodes, possibly as early as the LGM. The lake level highstands during the glacial period cannot be attributed to stadials or interstadials with absolute certainty due to the short-term nature of these events and uncertainties of the age-model during that period. Nevertheless, we hypothesize that similar mechanisms as in the late glacial (decreased evaporation during summers) also triggered late MIS 3/early MIS 2 lake transgressions.

On a spatial scale, most of the Central Asian mountains (i.e., the Pamirs, Hindukush, and Tien Shan) show enhanced summer precipitation during the glacial compared to the Holocene, except the eastern Pamirs



where this pattern is less pronounced due to the diminished influence of the westerlies (Figures 4b and S4). In contrast, PME is clearly higher during the LGM than during the Holocene throughout whole Central Asia (Figures 3 and 4c), highlighting the relevance of this parameter as major impacting factor on transregional hydroclimate.

4. Conclusions

 $\Delta_{terr-aq}$ and P_{aq} as integrative proxies for vapor source/winter westerlies intensity and catchment ecology/summer wetness, show distinct insolation-driven trends. δD -terr represents spring/summer δD -precip and its trend resembles those from other isotopic records from Asia, illustrating linkages between vapor-source changes over a large area. Based on our data it remains possible that the Indian summer monsoon touched the study area during early Holocene intensification. More likely, an intensified monsoon influenced the pathways and strengths of the westerlies, causing the most negative δD -terr values between 12 and 7 kyr BP.

 δ D-aq represents an integrated signal of the whole lake catchment. The large isotopic variability on glacialinterglacial scales can only be explained by significant changes in seasonality during the studied time period. Here, positive shifts at 31–30, 28–26, and 17–14 kyr BP are interpreted to be triggered by increased summer precipitation and/or low summer evaporation and decreased input from high-altitude meltwater. Synchronous higher concentrations of terrestrial biomarkers (lower P_{aq}) and pollen counts also give evidence for expansion of vegetation which reacts sensitively to slight increases in moisture availability in a hyperarid environment. These seasonality changes are in agreement with simulated data from CAM3-iso, ECHAM5-wiso and EC-Earth models and possibly constitute a local response to northern hemispheric rapid climate events.

Higher lake levels during these three intervals are supported by multiple indicators. For the late glacial, proxy evidence clearly indicate lake transgression during cold intervals, possibly starting as early as during LGM, long before a 14.7-kyr Bølling-warming and while glaciers were still expanding. Climate models point out decreased evaporation as a major factor that increases summer effective moisture availability and profoundly influences the hydrological states of lakes.

4.1. Author contributions

S. M. and Z. M. drilled the sediment core with local logistic support from I. R.; L. H. and M. K. conducted XRF scanning, sampled the core, and conducted δ^{18} O measurements of aragonite crystals. B. A. conducted organic geochemical analysis of samples in the labs of S. F. and D. S.; Q. Z., F. P., and M. W. conducted modelling experiments. B. A. wrote the manuscript with contributions from all co-authors.

Data Availability Statement

Data presented in this study are available on PANGAEA: https://doi.pangaea.de/10.1594/PANGAEA.907737

References

Aichner, B., Feakins, S. J., Lee, J. E., Herzschuh, U., & Liu, X. (2015). High-resolution leaf wax carbon and hydrogen isotopic record of the late Holocene paleoclimate in arid Central Asia. *Climate of the Past*, 11(4), 619–633.

- Aichner, B., Herzschuh, U., & Wilkes, H. (2010). Influence of aquatic macrophytes on the stable carbon isotopic signatures of sedimentary organic matter in lakes on the Tibetan Plateau. Organic Geochemistry, 41(7), 706–718.
- Aichner, B., Herzschuh, U., Wilkes, H., Schulz, H. M., Wang, Y. B., Plessen, B., et al. (2012). Ecological development of Lake Donggi Cona, north-eastern Tibetan Plateau, since the late glacial on basis of organic geochemical proxies and non-pollen palynomorphs. *Palaeogeography, Palaeoclimatology, Palaeoecology, 313*, 140–149.
- Aichner, B., Hilt, S., Perillon, C., Gillefalk, M., & Sachse, D. (2017). Biosynthetic hydrogen isotopic fractionation factors during lipid synthesis in submerged aquatic macrophytes: Effect of groundwater discharge and salinity. Organic Geochemistry, 113, 10–16.

Aizen, E. M., Aizen, V. B., Melack, J. M., Nakamura, T., & Ohta, T. (2001). Precipitation and atmospheric circulation patterns at midlatitudes of Asia. *International Journal of Climatology*, 21(5), 535–556.

Araguas-Araguas, L., Froehlich, K., & Rozanski, K. (1998). Stable isotope composition of precipitation over southeast Asia. Journal of Geophysical Research, 103(D22), 28,721–28,742.

Berger, A., & Loutre, M. F. (1991). Insolation values for the climate of the last 10 million years. *Quaternary Science Reviews*, 10(4), 297–317.
Bowen, G. J., & Revenaugh, J. (2003). Interpolating the isotopic composition of modern meteoric precipitation. *Water Resources Research*, 39(10), 1299. https://doi.org/10.1029/2003WR002086

Cai, Y. J., Chiang, J. C. H., Breitenbach, S. F. M., Tan, L. C., Cheng, H., Edwards, R. L., & An, Z. S. (2017). Holocene moisture changes in western China, Central Asia, inferred from stalagmites. *Quaternary Science Reviews*, 158, 15–28.

Acknowledgments We are grateful to K. Adler and T. Jonas

for help during coring and M. Rincon and M. Poehle for help during sample cleanup and δD analysis. Financial support to B. A. was provided by the German Academic Exchange Service (DAAD) in the form of two travel grants and by the German Science Foundation (DFG; project Ai 134/2-1 and 2-2). Q. Z. acknowledges support from the Swedish Research Council VR project 2013-06476. The simulations with CAM3 and EC-Earth were performed on resources provided by the Swedish National Infrastructure for Computing (SNIC) at NSC and Cray XC30 HPC systems at ECMWF. We acknowledge two anonymous reviewers whose comments improved the manuscript.



Caves Rugenstein, J. K., & Chamberlain, C. P. (2018). The evolution of hydroclimate in Asia over the Cenozoic: A stable-isotope perspective. Earth Science Reviews, 185, 1129–1156.

Chen, F. H., Chen, J. H., Holmes, J., Boomer, I., Austin, P., Gates, J. B., et al. (2010). Moisture changes over the last millennium in arid central Asia: A review, synthesis and comparison with monsoon region. *Quaternary Science Reviews*, 29(7-8), 1055–1068.

Cheng, H., Spötl, C., Breitenbach, S. F. M., Sinha, A., Wassenburg, J. A., Jochum, K. P., et al. (2016). Climate variations of Central Asia on orbital to millennial timescales. *Scientific Reports*, 6(1), 36975. https://doi.org/10.1038/srep36975

Chikaraishi, Y., Kaneko, M., & Ohkouchi, N. (2012). Stable hydrogen and carbon isotopic compositions of long-chain (C-21-C-33) *n*-alkanes and *n*-alkenes in insects. *Geochimica et Cosmochimica Acta*, 95, 53–62.

- Collins, W. D., Rasch, P. J., Boville, B. A., Hack, J. J., McCaa, J. R., Williamson, D. L., et al. (2006). The formulation and atmospheric simulation of the Community Atmosphere Model version 3 (CAM3). *Journal of Climate*, 19(11), 2144–2161.
- de Leeuw, J. W., van de Meer, F. W., Rijpstra, W. I. C., & Schenck, P. A. (1979). On the occurrence and structural identification of long chain unsaturated ketones and hydrocarbons in sediments. In *Advances in organic geochemistry* (pp. 211–217). Oxford: Pergamon Press.
- Deplazes, G., Luckge, A., Stuut, J. B. W., Patzold, J., Kuhlmann, H., Husson, D., et al. (2014). Weakening and strengthening of the Indian monsoon during Heinrich events and Dansgaard-Oeschger oscillations. *Paleoceanography*, 29, 99–114. https://doi.org/10.1002/ 2013PA002509
- Dissard, D., Nehrke, G., Reichart, G. J., & Bijma, J. (2010). The impact of salinity on the Mg/Ca and Sr/Ca ratio in the benthic foraminifera Ammonia tepida: Results from culture experiments. *Geochimica et Cosmochimica Acta*, 74(3), 928–940.

Dortch, J. M., Owen, L. A., & Caffee, M. W. (2013). Timing and climatic drivers for glaciation across semi-arid western Himalayan–Tibetan orogen. Quaternary Science Reviews, 78, 188–208.

Ficken, K. J., Li, B., Swain, D. L., & Eglinton, G. (2000). An n-alkane proxy for the sedimentary input of submerged/floating freshwater aquatic macrophytes. Organic Geochemistry, 31(7-8), 745–749.

Haggi, C., Eglinton, T. I., Zech, W., Sosin, P., & Zech, R. (2019). A 250 ka leaf-wax delta D record from a loess section in Darai Kalon, Southern Tajikistan. *Quaternary Science Reviews*, 208, 118–128.

Hazeleger, W., Severijns, C., Semmler, T., Ştefănescu, S., Yang, S., Wang, X., et al. (2010). EC-Earth: A seamless Earth-system prediction approach in action. Bulletin of the American Meteorological Society, 91(10), 1357–1364. https://doi.org/10.1175/2010BAMS2877.1

Hazeleger, W., Wang, X., Severijns, C., Ştefănescu, S., Bintanja, R., Sterl, A., et al. (2012). EC-Earth V2.2: Description and validation of a new seamless Earth system prediction model. *Climate Dynamics*, 39(11), 2611–2629.

Heinecke, L., Epp, L. S., Reschke, M., Stoof-Leichsenring, K. R., Mischke, S., Plessen, B., & Herzschuh, U. (2017). Aquatic macrophyte dynamics in Lake Karakul (Eastern Pamir) over the last 29 cal ka revealed by sedimentary ancient DNA and geochemical analyses of macrofossil remains. *Journal of Paleolimnology*, 58(3), 403–417.

Heinecke, L., Fletcher, W. J., Mischke, S., Tian, F., & Herzschuh, U. (2018). Vegetation change in the eastern Pamir Mountains, Tajikistan, inferred from Lake Karakul pollen spectra of the last 28 kyr. Palaeogeography, Palaeoclimatology, Palaeoecology, 511, 232–242.

Heinecke, L., Mischke, S., Adler, K., Barth, A., Biskaborn, B. K., Plessen, B., et al. (2017). Climatic and limnological changes at Lake Karakul (Tajikistan) during the last 29 cal ka. *Journal of Paleolimnology*, 58(3), 317–334.

Herzschuh, U. (2006). Palaeo-moisture evolution in monsoonal Central Asia during the last 50,000 years. *Quaternary Science Reviews*, 25(1-2), 163–178.

Hou, J. Z., D'Andrea, W. J., Wang, M. D., He, Y., & Liang, J. (2017). Influence of the Indian monsoon and the subtropical jet on climate change on the Tibetan Plateau since the late Pleistocene. *Quaternary Science Reviews*, 163, 84–94.

Jia, J., Liu, H., Gao, F. Y., & Xia, D. S. (2018). Variations in the westerlies in Central Asia since 16 ka recorded by a loess section from the Tien Shan Mountains. Palaeogeography, Palaeoclimatology, Palaeoecology, 504, 156–161.

Kageyama, M., D'Andrea, F., Ramstein, G., Valdes, P. J., & Vautard, R. (1999). Weather regimes in past climate atmospheric general circulation model simulations. *Climate Dynamics*, 15(10), 773–793.

Kageyama, M., Valdes, P. J., Ramstein, G., Hewitt, C., & Wyputta, U. (1999). Northern hemisphere storm tracks in present day and Last Glacial Maximum climate simulations: A comparison of the European PMIP models. *Journal of Climate*, *12*(3), 742–760.

Komatsu, T., & Tsukamoto, S. (2015). Late glacial lake-level changes in the Lake Karakul Basin (a Closed Glacierized-Basin), eastern Pamirs, Tajikistan. *Quaternary Research*, 83(1), 137–149.

Laine, A., Kageyama, M., Salas-Melia, D., Voldoire, A., Riviere, G., Ramstein, G., et al. (2009). Northern Hemisphere storm tracks during the last glacial maximum in the PMIP2 ocean-atmosphere coupled models: Energetic study, seasonal cycle, precipitation. *Climate Dynamics*, 32(5), 593–614.

Li, X., Liu, X., Qiu, L., An, Z., & Yin, Z.-Y. (2013). Transient simulation of orbital-scale precipitation variation in monsoonal East Asia and arid central Asia during the last 150 ka. *Journal of Geophysical Research: Atmospheres*, 118, 7481–7488. https://doi.org/10.1002/ jgrd.50611

Liu, H., Yang, H., Cao, Y., Leng, Q., & Liu, W. (2018). Inter-molecular variations of fatty acid δD in algae and submerged plants from the north-eastern Tibetan Plateau. Organic Geochemistry, 122, 17–28.

Liu, H., Yang, H., Cao, Y., & Liu, W. (2018). Compound-specific δD and its hydrological and environmental implication in the lakes on the Tibetan Plateau. Science China Earth Sciences, 61(6), 765–777.

Liu, W., Liu, Z., Fu, M., & An, Z. (2008). Distribution of the C37 tetra-unsaturated alkenone in Lake Qinghai, China: A potential lake salinity indicator. *Geochimica et Cosmochimica Acta*, 72(3), 988–997.

Liu, W., Liu, Z., Wang, H., He, Y., Wang, Z., & Xu, L. (2011). Salinity control on long-chain alkenone distributions in lake surface waters and sediments of the northern Qinghai-Tibetan Plateau, China. *Geochimica et Cosmochimica Acta*, 75(7), 1693–1703.

Mayer, L. M., Macko, S. A., Mook, W. H., & Murray, S. (1981). The distribution of bromine in coastal sediments and its use as a source indicator for organic matter. *Organic Geochemistry*, *3*(1), 37–42.

Miehe, G., Winiger, M., Böhner, J., & Zhang, Y. (2001). The climatic diagram map of High Asia. Purpose and concepts. *Erdkunde*, 55, 94–97.

Mischke, S., Lai, Z. P., Aichner, B., Heinecke, L., Mahmoudov, Z., Kuessner, M., & Herzschuh, U. (2017). Radiocarbon and optically stimulated luminescence dating of sediments from Lake Karakul, Tajikistan. *Quaternary Geochronology*, 41, 51–61.

Noone, D., & Sturm, C. (2010). Comprehensive dynamical models of global and regional water isotope distributions. In *Isoscapes:* Understanding movement, pattern, and process on Earth through isotope mapping (pp. 195–219). Netherlands: Springer.

Pausata, F. S. R., Battisti, D. S., Nisancioglu, K. H., & Bitz, C. M. (2011). Chinese stalagmite delta O-18 controlled by changes in the Indian monsoon during a simulated Heinrich event. *Nature Geoscience*, 4(7), 474–480.

Peterse, F., Prins, M. A., Beets, C. J., Troelstra, S. R., Zheng, H. B., Gu, Z. Y., et al. (2011). Decoupled warming and monsoon precipitation in East Asia over the last deglaciation. *Earth and Planetary Science Letters*, 301(1-2), 256–264.

- Rao, Z., Li, Y., Zhang, J., Jia, G., & Chen, F. (2016). Investigating the long-term palaeoclimatic controls on the δD and δ¹⁸O of precipitation during the Holocene in the Indian and East Asian monsoonal regions. *Earth Science Reviews*, 159, 292–305.
- Roeckner, E., Bäuml, G., Bonaventura, L., Brokopf, R., Esch, M., Giorgetta, M., et al. (2003), The general circulation model ECHAM5. Part I: Model description. Rep., Max Planck Institute for Meteorology, Hamburg.
- Roeckner, E., Brokopf, R., Esch, M., Giorgetta, M., Hagemann, S., Kornblueh, L., et al. (2006). Sensitivity of simulated climate to horizontal and vertical resolution in the ECHAM5 atmosphere model. *Journal of Climate*, 19(16), 3771–3791.

Rossby, C. G. (1939). Relation between variations in the intensity of the zonal circulation of the atmosphere and the displacements of the semi-permanent centers of action. Journal of Marine Research, 2, 38–55.

Sarkar, S., Prasad, S., Wilkes, H., Riedel, N., Stebich, M., Basavaiah, N., & Sachse, D. (2015). Monsoon source shifts during the drying mid-Holocene: Biomarker isotope based evidence from the core "monsoon zone" (CMZ) of India. Quaternary Science Reviews, 123, 144–157.

- Schiemann, R., Luthi, D., & Schar, C. (2009). Seasonality and interannual variability of the westerly jet in the Tibetan Plateau Region. *Journal of Climate*, 22(11), 2940–2957.
- Strecker, M. R., Frisch, W., Hamburger, M. W., Ratschbacher, L., Semiletkin, S., Samoruyev, A., & Sturchio, N. (1995). Quaternary deformation in the Eastern Pamirs, Tajikistan and Kyrgyzstan. *Tectonics*, 14(5), 1061–1079.

Sturm, C., Zhang, Q., & Noone, D. (2010). An introduction to stable water isotopes in climate models: Benefits of forward proxy modelling for paleoclimatology. *Climate of the Past*, 6(1), 115–129.

Taft, L., Mischke, S., Wiechert, U., Leipe, C., Rajabov, I., & Riedel, F. (2014). Sclerochronological oxygen and carbon isotope ratios in Radix (Gastropoda) shells indicate changes of glacial meltwater flux and temperature since 4,200 cal yr BP at Lake Karakul, eastern Pamirs (Tajikistan). Journal of Paleolimnology, 52(1-2), 27–41.

Tierney, J. E., Pausata, F. S. R., & deMenocal, P. (2016). Deglacial Indian monsoon failure and North Atlantic stadials linked by Indian Ocean surface cooling. *Nature Geoscience*, 9(1), 46–50.

- van Bree, L. G. J., Rijpstra, W. I. C., Cocquyt, C., Al-Dhabi, N. A., Verschuren, D., Damste, J. S. S., & de Leeuw, J. W. (2014). Origin and palaeoenvironmental significance of C₂₅ and C₂₇ n-alk-1-enes in a 25,000-year lake-sedimentary record from equatorial East Africa. *Geochimica et Cosmochimica Acta*, 145, 89–102.
- Wang, P. X., Wang, B., Cheng, H., Fasullo, J., Guo, Z. T., Kiefer, T., & Liu, Z. Y. (2017). The global monsoon across time scales: Mechanisms and outstanding issues. *Earth Science Reviews*, 174, 84–121.
- Wang, Y. J., Cheng, H., Edwards, R. L., An, Z. S., Wu, J. Y., Shen, C.-C., & Dorale, J. A. (2001). A High-Resolution Absolute-Dated Late Pleistocene Monsoon Record from Hulu Cave. *China. Science*, 294, 2345–2348. https://doi.org/10.1126/science.1064618

Wang, Y. J., Cheng, H., Edwards, R. L., He, Y. Q., Kong, X. G., An, Z. S., et al. (2005). The Holocene Asian monsoon: Links to solar changes and North Atlantic climate. Science, 308(5723), 854–857. https://doi.org/10.1126/science.1106296

Werner, M., Langebroek, P. M., Carlsen, T., Herold, M., & Lohmann, G. (2011). Stable water isotopes in the ECHAM5 general circulation model: Toward high-resolution isotope modeling on a global scale. *Journal of Geophysical Research*, 116, D15109. https://doi.org/ 10.1029/2011JD015681

Williams, M. W., & Konovalov, V. G. (2008). Central Asia temperature and precipitation data, 1879–2003, version 1., edited, NSIDC: National Snow and Ice Data Center Boulder, Colorado, USA.

- Yan, D., & Wünnemann, B. (2014). Late Quaternary water depth changes in Hala Lake, northeastern Tibetan Plateau, derived from ostracod assemblages and sediment properties in multiple sediment records. *Quaternary Science Reviews*, 95, 95–114.
- Yan, D. N., Xu, H., Lan, J. H., Zhou, K. G., Ye, Y. D., Zhang, J. X., et al. (2019). Solar activity and the westerlies dominate decadal hydroclimatic changes over arid Central Asia. Global and Planetary Change, 173, 53–60. https://doi.org/10.1016/j.gloplacha.2018.12.006
- Yao, T., Masson-Delmotte, V., Gao, J., Yu, W., Yang, X., Risi, C., et al. (2013). A review of climatic controls on delta O-18 in precipitation over the Tibetan Plateau: Observations and simulations. *Reviews of Geophysics*, 51, 525–548. https://doi.org/10.1002/rog.20023

Zhang, X. J., Jin, L. Y., Chen, J., Chen, F. H., Park, W., Schneider, B., & Latif, M. (2017). Detecting the relationship between moisture changes in arid central Asia and East Asia during the Holocene by model-proxy comparison. *Quaternary Science Reviews*, 176, 36–50.
Zhang, X. J., Li, J. R., Gao, M., & Jin, L. Y. (2018). Simulated precipitation changes in Central Asia since the Last Glacial Maximum.

Quaternary International, 490, 82–97.

Zhang, Y. D., Su, Y. L., Liu, Z. W., Chen, X. C., Yu, J. L., Di, X. D., & Jin, M. (2015). Long-chain *n*-alkenes in recent sediment of Lake Lugu (SW China) and their ecological implications. *Limnologica*, *52*, 30–40.